# Impact of bottom trawling on long-term carbon sequestration in shelf sea sediments

Wenyan Zhang<sup>1</sup>, Lucas Porz<sup>1</sup>, Rümeysa Yilmaz<sup>1</sup>, Jannis Kuhlmann<sup>2</sup>, Andreas Neumann<sup>3</sup>, Bo Liu<sup>4</sup>, Daniel Müller<sup>5</sup>, Timo Spiegel<sup>6</sup>, Moritz Holtappels<sup>7</sup>, Nadja Ziebarth<sup>2</sup>, Bettins Taylor<sup>2</sup>, Klaus Wallmann<sup>8</sup>, Sabine Kasten<sup>7</sup>, Ute Daewel<sup>9</sup>, and Corinna Schrum<sup>10</sup>

<sup>1</sup>Institute of Coastal Systems-Analysis and Modeling, Helmholtz-Zentrum hereon
<sup>2</sup>BUND-Meeresschutzbüro, Bund für Umwelt und Naturschutz Deutschland e.V. (BUND)
<sup>3</sup>Helmholtz-Zentrum hereon, Institute of Carbon Cycles
<sup>4</sup>Alfred Wegener Institute
<sup>5</sup>Alfred Wegener Institute Helmholtz Centre for Polar and Marine Research
<sup>6</sup>GEOMAR Helmholtz Centre for Ocean Research Kiel
<sup>7</sup>Alfred Wegener Institute for Polar and Marine Research
<sup>8</sup>GEOMAR
<sup>9</sup>Institute of Coastal Research, Helmholtz-Zentrum Geesthacht
<sup>10</sup>University of Hamburg, Institute of Oceanography

February 7, 2023

#### Abstract

Bottom trawling represents the most widespread anthropogenic physical disturbance to shelf sea sediments. While trawlinginduced mortality in benthic fauna has been extensively investigated, its impacts on ecosystem functioning and carbon cycling at regional scales remain unclear. Using the North Sea as an example, we address these issues by synthesizing a high-resolution dataset of bottom trawling impact on sediments, feeding this dataset into a 3-dimensional physical-biogeochemical model to estimate trawling-induced changes in biomass, bioturbation and sedimentary organic carbon, and assessing model results with field samples. Results suggest a trawling-induced net reduction in macrobenthic biomass by 10-27%. Trawling-induced resuspension and reduction of bioturbation jointly and accumulatively reduce the regional sedimentary organic carbon sequestration capacity by 21-67%, equivalent to 0.58-1.84 Mt CO2 yr-1. Our study emphasizes the need for proper management of trawling on muddy seabeds, if the natural capacity of shelf seas for carbon sequestration should be conserved and restored.

#### Hosted file

954687\_0\_art\_file\_10676631\_rpldtx.docx available at https://authorea.com/users/536828/ articles/623152-impact-of-bottom-trawling-on-long-term-carbon-sequestration-in-shelfsea-sediments

#### Hosted file

954687\_0\_supp\_10676632\_rpmm4m.docx available at https://authorea.com/users/536828/articles/ 623152-impact-of-bottom-trawling-on-long-term-carbon-sequestration-in-shelf-seasediments

- Impact of bottom trawling on long-term carbon sequestration in shelf sea sediments 2
- W. Zhang<sup>1</sup>, L. Porz<sup>1</sup>, R. Yilmaz<sup>1</sup>, J. Kuhlmann<sup>2</sup>, A. Neumann<sup>3</sup>, B. Liu<sup>4</sup>, D. Müller<sup>4</sup>, T. Spiegel<sup>5</sup>, M. Holtappels<sup>4, 8</sup>, N. Ziebarth<sup>2</sup>, B. Taylor<sup>2</sup>, K. Wallmann<sup>5</sup>, S. Kasten<sup>4, 7, 8</sup>, U. Daewel<sup>1</sup> and C. Schrum<sup>1,6</sup> 3
- 4
- 5
- <sup>1</sup> Institute of Coastal Systems-Analysis and Modeling, Helmholtz-Zentrum Hereon, Geesthacht, 6
- Germany 7
- <sup>2</sup> BUND-Meeresschutzbüro, Bund für Umwelt und Naturschutz Deutschland e.V. (BUND), 8
- Bremen, Germany 9
- <sup>3</sup> Institute of Carbon Cycles, Helmholtz-Zentrum Hereon, Geesthacht, Germany 10
- <sup>4</sup> Alfred Wegener Institute Helmholtz Centre for Polar and Marine Research, Bremerhaven, 11
- Germany 12
- <sup>5</sup>GEOMAR-Helmholtz Centre for Ocean Research Kiel, Kiel, Germany 13
- <sup>6</sup> Institute of Oceanography, Center for Earth System Research and Sustainability, University of 14
- Hamburg, Hamburg, Germany 15
- <sup>7</sup> Faculty of Geosciences, University of Bremen, Bremen, Germany 16
- <sup>8</sup> Center for Marine Environmental Sciences MARUM, University of Bremen, Bremen, 17
- Germany 18
- 19
- Corresponding author: Wenyan Zhang (wenyan.zhang@hereon.de) 20

#### **Key Points:** 21

- Not only the intensity but also the temporal variability of bottom trawling has long-22 lasting impact on carbon sequestration 23
- Trawling-induced resuspension and -impaired bioturbation jointly reduce the regional • 24 organic carbon sequestration capacity by 21-67% 25
- Organic carbon stored in the muddy seabed is most sensitive to disturbance by bottom 26 trawling 27

## 28 Abstract

- 29 Bottom trawling represents the most widespread anthropogenic physical disturbance to shelf sea
- 30 sediments. While trawling-induced mortality in benthic fauna has been extensively investigated,
- 31 its impacts on ecosystem functioning and carbon cycling at regional scales remain unclear. Using
- 32 the North Sea as an example, we address these issues by synthesizing a high-resolution dataset of
- bottom trawling impact on sediments, feeding this dataset into a 3-dimensional physical–
- biogeochemical model to estimate trawling-induced changes in biomass, bioturbation and
- 35 sedimentary organic carbon, and assessing model results with field samples. Results suggest a
- trawling-induced net reduction in macrobenthic biomass by 10-27%. Trawling-induced resuspension and reduction of bioturbation jointly and accumulatively reduce the regional
- sedimentary organic carbon sequestration capacity by 21-67%, equivalent to 0.58-1.84 Mt CO<sub>2</sub>
- $yr^{-1}$ . Our study emphasizes the need for proper management of trawling on muddy seabeds, if the
- 40 natural capacity of shelf seas for carbon sequestration should be conserved and restored.
- 41

## 42 Plain Language Summary

Ocean seabed has been subjected to disturbance by bottom trawling for quite a long time. A 43 bottom trawl consists of a large conic net with a wide mouth and a small enclosed end. The 44 mouth is usually equipped with heavy components to keep the net open and attached on the 45 seabed, effectively crushing every small obstacle in its path. Because of this physical contact, 46 seabed sediments are massively resuspended and mixed, and animals living on or in the 47 sediments are caught, killed or injured. Despite the damage to seabed habitats, quantitative 48 assessment of the impact of trawling on ecosystem and carbon cycling remains poorly 49 constrained. Our study presents an effort towards a better quantitative assessment by 50 constructing a high-resolution, long-term dataset of trawling impacts on sediments, analyzing 51 sediment samples, and performing elaborate modeling. Our results show that bottom trawling not 52 only affects the animals and their function of carbon reworking, but also has a cumulative impact 53 on long-term carbon storage. An almost 20-year reduction of bottom trawling effort has allowed 54 the seabed animals to recover to some extent, but it was insufficient to restore the amount of 55

- <sup>56</sup> organic carbon deposited in the sediment, especially in the trawled muddy areas.
- 57

# 58 **1 Introduction**

- 59 Bottom trawling represents the most widespread anthropogenic physical disturbance to surface
- 60 sediments and benthic habitats (Halpern et al., 2008; Sala et al., 2021). A recent estimate
- suggests that ~22 Gt of sediment are resuspended by bottom trawling every year on the world's
- 62 continental shelves, comparable to the total annual sediment supply through rivers (Oberle et al.,
- 63 2016). Although bottom trawling has been performed for over a century on heavily fished
- 64 grounds such as the North Sea, quantifying its impact on benthic biota and biogeochemical
- cycling has become a focal point of research only in recent decades (Halpern et al., 2008;
- 66 Hiddink et al., 2017; Amoroso et al., 2018; Sala et al., 2021). In particular, the combination of
- 67 satellite-based monitoring systems and logbook data have enabled estimates of bottom trawling
- 68 with high spatial and temporal resolution (Eigaard et al., 2017; Kroodsma et al., 2018; Pitcher et
- 69 al., 2022).

- 70 Bottom trawling significantly restructures the top layers of seafloor sediments and thereby alters
- <sup>71</sup> benthic faunal communities (Hiddink et al., 2017; Tiano et al., 2022), biogeochemical processes
- and element fluxes across the sediment-water interface (De Borger et al., 2021). A recent study
- by Sala et al. (2021) highlighted the impact of bottom trawling on the storage of organic carbon
- 74 (OC) in seafloor sediments. They suggested that trawling causes remineralization of 0.16-0.4 Gt
- of sedimentary OC globally every year, resulting in an emission of 0.58-1.47 Gt of CO<sub>2</sub>.
- <sup>76</sup> However, this estimate was subsequently questioned by Epstein et al (2022), who pointed out
- <sup>77</sup> large uncertainties in several assumptions made by Sala et al. (2021). Epstein et al. (2022)
- remphasized that there is no consensus in existing literature on whether or to what extent trawling
- 79 would lead to reduced OC storage. Understanding the fate of OC disturbed by trawling in 80 different environmental settings was identified as a key research gap that would allow
- incorporating evidence-based considerations of OC sequestration into seabed management.
- The blurred picture of the net effect of bottom trawling on OC sequestration in sediments originates from several counteracting mechanisms. On the one hand, trawling-induced
- resuspension and physical mixing of sediments may enhance remineralization of OC by
- increasing oxygen exposure time and limit OC sequestration by inhibiting deposition and burial
- of fine-grained sediments (Hartnett et al., 1998; Zonneveld et al., 2010; Keil, 2017; Freitas et al.,
- 2021; Paradis et al., 2021). On the other hand, intensified lateral transport towards offshore
- depocenters and increase in primary production from the resuspension of nutrients may offset the
- loss of OC to varying extents (Dounas et al., 2007; Martín et al., 2008; Paradis et al., 2018). The
- complexity of OC sequestration is further increased by benthic fauna, which do not only
- 91 contribute OC but also play a vital role in mediating OC fluxes across the sediment-water
- interface (Middelburg, 2018; Zhang et al., 2021). The overall effect of trawling on macrobenthos
- is a depletion of biomass and a change of the community from sessile to mobile and
- 94 opportunistic species (Kaiser et al., 2006; Sciberras et al., 2018; Tillin et al., 2006). However, for
- 95 quantitative assessments of how trawling-induced impacts on benthic fauna would affect OC
- sequestration, the current evidence base is incomplete and often contested (Epstein et al., 2022;
- 97 LaRowe et al., 2020).
- As a heavily fished region, the North Sea has been subject to trawling for more than one century
- 99 (Thurstan et al., 2010). Bottom trawling is performed in ~90% of the North Sea area (ICES,
- 100 2021), with a particularly high intensity in shallow waters (water depth < 300 m) of the
- 101 Skagerrak, where the swept area ratio (SAR) reaches  $\geq 10 \text{ yr}^{-1}$  (Figure 1). Bottom trawling
- intensity shows a positive correlation (r = 0.18, p < 0.001) with OC contents in near-surface
- 103 sediments, implying a disproportionately high trawling impact in areas of natural OC burial
- 104 (Figure 1). Existing estimates of the total stock of OC in the uppermost 10 cm of North Sea
- sediments range between 96 and 476 Mt (Bockelmann et al., 2018; Wilson et al., 2018; Diesing
- et al., 2021), with a major part (~75%) being stored in mud depocenters which represent ~20% of
- 107 the North Sea area (Figure 1). Previous studies have quantified the oceanic and atmospheric OC
- loads as well as OC accumulation rates (de Haas et al. 1997; Thomas et al., 2005; Diesing et al.,
- 109 2021), and identified the footprint of trawling on benthic habitats and ecosystem (Eigaard et al.,
- 110 2017; Rijnsdorp et al., 2018). However, no study has yet addressed the impact of bottom
- 111 trawling on the temporal and spatial variability of OC storage capacity in the North Sea (Diesing
- et al., 2021). To achieve this objective, we have synthesized a high-resolution, long-term (71
- 113 years) dataset of bottom trawling impacts on sediments, applied 3-dimensional physical-
- biogeochemical modeling based on the dataset to estimate trawling-induced change of
- sedimentary OC stock, and evaluated model results with field samples.



Figure 1. Top: (a) Mean mud content (%) in the upper 10 cm sediment, with marked large-scale mud depocenters. (b) Mean Total Organic Carbon (TOC) content (%) in surface 10 cm sediment interpolated from existing field data. Source of (a) & (b): Bockelmann et al. (2018). Bottom:
Synthesized annual total bottom trawling intensity in terms of swept area ratio (SAR, yr<sup>-1</sup>) averaged over the period of 2015-2020 in (c) surface sediments (0-2 cm) and (d) sub-surface sediments (2-10 cm). The contour line of SAR=1 is indicated in (c) and (d). All plots are in logarithmic scale.

124

#### 125 2 Materials and Methods

#### 126 2.1 Synthesis of Bottom Trawling Data 1950-2020

127 Annually aggregated spatial data on bottom trawling in the North Sea including gear-type

information for 2009-2017 (ICES, 2019) were combined with daily fishing effort data for 2012-

129 2020 (Kroodsma et al., 2018) in order to generate a daily time-series of fishing effort on a

- $130 \quad 0.1^{\circ} \times 0.1^{\circ}$  grid. Based on the penetration depths of different gear components in different
- sediment types, the daily SAR fields in three depth intervals in seafloor sediment (0-2 cm, 2-5 cm, 2-5
- 132 cm, and 5-10 cm) were generated (Text S1).

A historical reconstruction of bottom trawling effort (Couce et al., 2020) was used to hindcast 133 trawling back until 1985. The availability of bottom trawling data before 1985 is, however, 134 135 limited. Two methods were employed to fill the data gap for the period 1950-1984. Method 1 is based on the relationship between trawling effort and total annual landings. The time series 136 1985-2020 of bottom trawling effort shows a high correlation (r = 0.86) with the annual landing 137 data of demersal and benthic fish (ICES, 2021; Figure S1). Based on this correlation, we 138 extrapolated the time series of bottom trawling data back to 1950 through a linear regression 139 between the trawling and landing data. Method 2 is based on the relationship between the total 140 trawling effort and the UK portion in the North Sea. Among all North Sea countries, trawling 141 data from the UK have the longest temporal coverage dating back to the 1910s (Engelhard, 2016). 142 The trawling effort of UK vessels shows an almost linear (r = 0.98) relationship with the total 143 effort in the North Sea in 2003-2020. Based on this relationship, the total trawling effort for 144 1950-1984 was assumed to also scale linearly with the UK trawling effort. Results from the two 145 reconstructions show distinct variations (Figure S1). Method 1 yields a high fishing effort in the 146 1970s and 1980s that is comparable to that of the 1990s and a lower level in the 1950s and 1960s, 147 148 comparable to the 2010s, whilst Method 2 produces almost the opposite pattern. Both reconstructed datasets were used in simulation to constrain the uncertainty in the estimation. The 149 spatial and seasonal distribution of trawling effort in the time span 1950-2015 is assumed to 150

follow the same pattern as in 2016-2020 based on the fact that the distribution of fishing effort

among countries, seasons and gear types is fairly stable (Couce et al., 2020, Figure S2&S3).

## 153 2.2 Numerical Modeling

We applied a 3-dimensional physical-biogeochemical model to simulate the impact of bottom 154 trawling on macrobenthos and OC cycling in surface sediments (Figure 2). The model consists of 155 three major components, with HAMSOM (HAMburg Shelf Ocean Model, Schrum, 1997) for 156 hydrodynamics, ECOSMO (ECOSystem Model, Daewel & Schrum, 2013) for biogeochemistry 157 and TOCMAIM (TOC-MAcrobenthos Interaction Model, Zhang & Wirtz, 2017) for 158 macrobenthos. OC is divided into three pools depending on the degradability, namely labile (i.e., 159 of high nutritional quality for macrobenthos), semi-labile (i.e., of intermediate nutritional 160 quality), and refractory (i.e., of low nutritional quality). With a Neumann boundary condition of 161 OC fluxes at the sediment-water interface calculated by the pelagic model component, 162 sedimentary OC content is solved by a mass balance equation including the impacts of 163 deposition/erosion, oxygen-dependent first-order degradation, macrobenthic uptake and 164 bioturbation. Temporal change of macrobenthic biomass is calculated based on food availability 165 in the form of OC, temperature, oxygen, and mortality caused by predation and bottom trawling. 166 Bioturbation scales with macrobenthic biomass and is inversely related with local OC resource. 167 The impact of bottom trawling on OC sequestration in sediments is implemented through 168 enhanced OC resuspension and enhanced mortality of macrobenthos. The latter subsequently 169 affects macrobenthic uptake of OC and bioturbation intensity, which feed back to the vertical 170 mixing efficiency, and thereby to OC sequestration. Based on field assessments of trawling-171 induced mortality of benthic fauna (Bergman & Van Santbrink, 2000; Hiddink et al., 2017), three 172 trawling impact scenarios, namely Trawling 10th pct, Trawling 50th pct and Trawling 90th pct 173 representing low (10th percentile, corresponding to depletion rate of 0.11 for SAR=1), medium 174

175 (50th percentile / depletion rate of 0.2) and high (90th percentile / depletion rate of 0.3) impact

levels in mortality, respectively, were simulated and compared to a non-trawling scenario (No-trawling). Model details are provided in Text S2.

178



179

Figure 2. Schematic illustrating the benthic-pelagic coupling and impact of bottom trawling
 implemented in the HAMSOM-ECOSMO-TOCMAIM coupled numerical model.

## 182 2.3 Sediment Sampling and Analysis

183 Our simulation results have been confirmed by field data of OC and macrobenthic biomass (in

ash free dry weight hereafter) in seafloor surface sediments of the southern North Sea (Zhang et
al., 2019). In this study, the model domain was extended to the entire North Sea and additional

field data from the northern part (Table S3) were compiled to assess the model performance

(Figure 3). In particular, recently-collected sediment samples in both trawled and untrawled areas
 were analyzed. Processing procedures are the same as described in Zhang et al. (2019) and

were analyzed. Processing procedures are the same as described in Zhang et al. (2019) and
 explained further in Text S3. The measured OC data and estimated bioturbation rates at these

field stations were compared to the simulation results (Figure S5).

# 191 **3 Results and Discussion**

# 192 3.1 Impact of trawling on macrobenthos and bioturbation

The long-term hindcast simulation encompasses the period 1950 - 2016. The simulation period for model spinup until approaching a relatively stable level for macrobenthic biomass is about 25

vears (Zhang et al., 2019). We focus on analysis of results starting from 1985, after which the

time series of trawling are identical for both methods of reconstruction (section 2.1). The

197 simulation results show similar variation patterns for 1985-2016 using both methods (Figure 3).

198 Comparison with field data suggests a generally satisfactory model performance in capturing

199 spatial distribution of macrobenthic biomass (Figure 3a, b), TOC content and bioturbation rates

200 (Figure S5). In all scenarios, simulation results confirm a peak of spatial mean biomass in the

area between 53° and 54°N, and a decrease toward the north in the shelf (water depth < 300 m)

- as reported by ICES (2007). The mean biomass rises again at  $\sim$ 57.5° owing to increased
- production along the shelf edge and shallow parts (water depth < 300 m) of the Skagerrak, with
- reported field measurements between 5 to 30 g m<sup>-2</sup> by Rosenberg et al. (1996) and Ståhl et al.
- (2004), and model results between 3 and 26 g m<sup>-2</sup>. Another peak in mean biomass occurs in the area between 58° and 59°N, where two large-scale mud depocenters (the Fladen Ground and the
- Norwegian Trench) are located (Figure 1a). Reported field data of macrobenthic biomass give 11
- $g \text{ m}^{-2}$  (spatial mean) in the Fladen Ground (de Wilde & Kok, 1986) and between 2-5 g m<sup>-2</sup> in the
- deep trench (Rosenberg et al., 1996). Our simulation results show consistency with field data in
- the Fladen Ground, whilst the values in the deep trench seem to be overestimated (Figure 4a).
- The long-term time series of modelled total biomass for the entire North Sea in each scenario
- show a generally stable level between 1985 and 2016 characterized mainly by seasonal
- 213 fluctuations associated with food supply (Figure 3c, d). The measured values from two large-
- scale ICES surveys (April-May in 1986 and May-June in 2000) are approximately the same
- 215 (3.06 Mt, Duineveld et al., 1991; ICES, 2007) and lie between the results of the scenarios
- Trawling\_10th pct and Trawling\_50th pct, about 20% lower than in the No-trawling scenario.
- 217 Seasonal fluctuations are normally between 0.15 0.6 Mt, accounting for 5-15% of the total
- biomass. Besides seasonal fluctuations, inter-annual variations ( $\leq 0.3$  Mt yr<sup>-1</sup>) and longer-term
- variations are also seen. The former is mainly related to hydro-climatic fluctuations (reflected by
- 220 the North Atlantic Oscillation winter index, see Zhang et al. (2019)) impacting the primary
- production (Figure 3f), whilst the latter appears to be associated with variations in the trawling
- effort, e.g. an increasing trend of biomass since 2008 due to reduced trawling (Figure 3c,d).
- For the period 1985 2008, the trawling-induced reduction of biomass ranges between 14-17%
- in the low-impact scenario (Trawling\_10th pct) and between 23-27% in the high-impact scenario
- 225 (Trawling\_90th pct) compared to the No-trawling scenario when using the reconstructed
- trawling data from Method 1. Biomass reduction is lower when using the reconstructed data
- from Method 2, with 10-14% in the low-impact scenario and 21-24% in the high-impact scenario,
- respectively. In all trawling scenarios, an apparent increase of biomass is seen starting in 2008
- 229 (Figure 3c,d). The biomass reduction amounts to 11% (Trawling\_10th pct) and 17%
- 230 (Trawling\_90th pct) in 2015-2016 using reconstructed data from Method 1. Similar values are
- seen in the results using reconstructed data from Method 2. This trend is caused by a decline of
- bottom-trawling starting in 2001, with the total annual effort dropping from  $\sim 3 \times 10^6$  hours in 2001 to  $\sim 1.6 \times 10^6$  hours in 2008 and remaining relatively stable afterwards (Figure 3c,d). It is
- 233 2001 to  $\sim 1.6 \times 10^6$  hours in 2008 and remaining relatively stable afterwards (Figure 3c,d). 234 worth noting that although the total accumulative trawling effort is  $\sim 10\%$  higher in the
- reconstructed time series (1950-1984) from Method 2 than that from Method 1 (Figure S1),
- simulation results indicate higher biomass during 1985 2008 using Method 2 (as described
- above). This is due to lower trawling effort in the reconstructed time series from Method 2 for
- 1968 1984, which allows a recovery of biomass that is similar to the pattern in 2008-2016. The
- difference between the two results becomes smaller with time but still exists until 2016,
- suggesting a long-lasting impact of historical trawling on macrobenthos.
- 241 The reduction in biomass obtained from our simulations is consistent with the estimates by
- Rijnsdorp et al. (2020), who combined information about trawling pressure, gear characteristics,
- habitat characteristics and sensitivity of benthic community to assess impact of bottom trawling
- for the period 2010–2012. Their results suggest a 13% reduction of total biomass in the North
- Sea by trawling, with highest reduction (26%) in fine-grained sediments (mud) and lowest
- reduction (9%) in coarse sediments. Similar patterns are also seen in our simulation results, with

- biomass reduction of up to 30% in frequently trawled (annual SAR > 5) mud depocenters (e.g.
- the Fladen Ground and the Skagerrak) and up to 15% in trawled sandy seabeds (Figure 4a, b).
- 249 Meanwhile, some un-trawled and less trawled areas are featured by a slight increase of biomass
- (< 5 %) due to deposition of resuspended OC from adjacent trawled areas.
- At a regional scale, the impact of bottom trawling on macrobenthos is not merely a reduction of
- biomass, which is within 30% in existing estimations (Hiddink et al., 2017; Rijnsdorp et al., 2020)
- including this study. Biomass reduction together with trawling-induced changes of benthic
- community structure may lead to altered dynamics in benthic carbon cycling (Paradis et al.,
   2021; Sciberras et al., 2018). Bioturbation as a key mechanism for transporting OC from surface
- sediments to deeper horizons is significantly impaired, and is only partly counterbalanced by
- trawling-induced physical mixing in the upper most few centimeters (De Borger et al., 2021).
- 258 Our simulation results (Figure 4c, d) suggest a substantial trawling-induced reduction of
- bioturbation rates in the North Sea, with largest reduction (up to 50%) in frequently trawled
- 260 (annual SAR > 5) muddy seafloor, and moderate reduction (10-20%) in frequently trawled
- 261 (annual SAR > 3) sandy seafloor. Meanwhile, the edges of frequently trawled areas feature a
- slight increase of bioturbation due to enhanced OC deposition, though these parts are still subject
- to trawling but with a much lower intensity (annual SAR < 0.1, Figure 1c, d). It is worth noting
- that the areas displaying the largest biomass reduction are also hotspots of bioturbation in the
- No-trawling scenario (Figure 4c). In energetic coastal seas such as the North Sea, where natural
- sedimentation rates are generally low (<  $1 \text{ mm yr}^{-1}$ , Diesing et al., 2021), bioturbation plays a
- key role in transporting OC from the oxic sediment surface to deeper anoxic horizons, thereby
- facilitating long-term OC sequestration (Zhang et al., 2019). The reduction of bioturbation in
- these key functional hotspots of OC sequestration would inevitably have a long-term impact.
   Though ecosystem changes in the North Sea have been attributed mainly to variations in climate
- and nutrients (Clark & Frid, 2001), our results point toward trawling as another possible main
- 272 driver of changes in benthic ecosystem.





Figure 3. (a) Scatter plot showing a general consistency between simulated macrobenthic biomass (scenario: Trawling\_50th pct from Method 1) and station data. Annual SAR is used to distinguish trawled (SAR >= 1) and non-trawled (SAR < 1) stations. (b) Distribution of mean macrobenthic biomass for each degree of latitude in the year 2000. Model scenarios are based on

Method 1-reconstructed trawling data. (c) Time series of simulated total macrobenthic biomass in the North Sea using Method 1-reconstructed data (gray line with y-axis on the right), with two large-scale ICES survey results marked (ICES, 2007). (d) Similar to (c) but using Method 2-

- reconstructed data. (e) Time series of estimated net change of total organic carbon (TOC) stock
- in the North Sea sediments in four model scenarios using Method 1-reconstructed data. The
- reference is the initial TOC stock (Figure 1b) specified for the simulation. (f) Similar to (e) but using Method 2-reconstructed data. Time series of simulated net annual primary production is
- 284 285
- indicated by the grey line with y-axis on the right in (f).

## 286 3.2 Impact of trawling on OC sequestration

Organic carbon preserved in surface sediments is susceptible to the impact of trawling. In 287 particular, the labile fraction is largely depleted in frequently trawled sites (De Borger et al., 288 2021; Paradis et al., 2021). Our simulation results show that the natural sequestration capacity of 289 290 OC in North Sea sediments, reflected by a net change of the OC stock, amounts to 0.2-1.2 Mt yr <sup>1</sup> with a long-term mean of 0.75 Mt  $vr^{-1}$  in the No-trawling scenario (Figure 3e, f). Seasonal 291 fluctuations associated with deposition of planktonic detritus range between 1 and 3 Mt. The 292 293 ratio of the annual net increase to the seasonal deposition suggests that 10-25% of deposited OC would be preserved without trawling. This natural sequestration capacity is increasingly impaired 294 with an increasing trawling-induced mortality in macrobenthos (Figure 3e, f). In the low-impact 295 scenarios (Trawling 10th pct), the annual net changes in the OC stock range between -0.05 and 296 1.25 Mt yr<sup>-1</sup> with a long-term mean accumulation of 0.6 Mt yr<sup>-1</sup>, corresponding to a 20% 297 reduction of the natural sequestration capacity. This reduction reaches 67% (using Method 1-298 299 reconstructed data) and 60% (using Method 2-reconstructed data) in the high-impact scenarios (Trawling 90th pct). Similar to the spatial patterns in the change of biomass and bioturbation, 300 frequently trawled muddy areas show the largest reduction of OC (10-20%) compared to the No-301 trawling scenario, while frequently trawled sandy areas feature a 5-10% reduction of OC (Figure 302 4e, f). Meanwhile, untrawled and less trawled areas (annual SAR < 0.1) are characterized by a 303 slight increase of sedimentary OC ( $\leq 10\%$ ), but this increase can offset the net decrease only to a 304 small extent (Figure 3e, f). 305

- 306 Our simulated OC accumulation rates in the North Sea are lower but still at the same order of
- magnitude as previous estimates, e.g. 1.1 Mt yr<sup>-1</sup> by de Haas et al., (1997) and  $1.43 \pm 2.07$  Mt yr<sup>-1</sup>
- by Diesing et al. (2021), and confirm the Norwegian Trough and Skagerrak as the major OC
- depocenters. Trawling impacts were not explicitly distinguished in previous estimates, which
   were based on interpretation of sediment samples. The disentangling of trawling impacts in our
- were based on interpretation of sediment samples. The disentangling of trawling impacts in our simulations allows an assessment of modified OC sequestration capacity in the depocenters. A
- comparison between the long-term varying trends of macrobenthic biomass and the sedimentary
- 313 OC stock (Figure 3) shows that in contrast to the (albeit slow) recovery of biomass due to
- reduction of trawling in the past decade, the divergence in the OC stock between the trawling
- and No-trawling scenarios becomes increasingly larger. This indicates that although an almost
- 316 20-year reduction of bottom trawling could allow the stock of macrobenthic biomass to recover
- to some extent, it is insufficient to restore the OC sequestration rates, especially in the trawled
- 318 muddy areas. Indeed, the muddy areas still show a reduced capacity for OC sequestration due to
- a persistently high trawling pressure there.
- 320 It is noteworthy that the North Sea still acts as a net carbon sink, albeit with strongly reduced
- 321 capacity by trawling, as simulated in the high-impact scenario. For that scenario, the spatially

- averaged reduction of OC storage capacity over the entire North Sea amounts to  $0.87 \text{ t C km}^{-2} \text{yr}^{-1}$ ,
- equivalent to 3.19 t CO<sub>2</sub> km<sup>-2</sup>yr<sup>-1</sup>. This is much lower than the global estimate of Sala et al.
- (2021), who presented an average steady-state remineralization in trawled areas of 32 t C km<sup>-2</sup>yr<sup>-1</sup>
- <sup>1</sup> or 118 t  $CO_2$  km<sup>-2</sup>yr<sup>-1</sup>. The difference between these estimates is likely explained by the simplified model of trawling impacts in Sala et al. (2021), which did not include natural
- remineralization and benthic ecosystem functioning. To further constrain the uncertainty in the
- estimate, other relevant processes in OC cycling and sequestration (e.g. gear-induced mixing and
- nutrient resuspension, bedform migration) need to be incorporated. Nevertheless, our study
- points out a central role of benthic ecosystem functioning in long-term carbon sequestration in
- 331 shelf seas.



332

Figure 4. Simulation results of the No-trawling scenario in August 2015 and trawling-induced
accumulative relative changes (scenario: Trawling\_50th pct) in August 2015. (a) and (b) are for
macrobenthic biomass. (c) and (d) are for mean bioturbation rates in the upper-most 10 cm
sediments. (e) and (f) are for mean TOC content in the upper-most 10 cm sediment. Positive and
negative values in (b), (d) and (f) indicate increase and reduction, respectively.

#### 339 Acknowledgments

- 340 This study is a contribution to the BMBF MARE:N project "Anthropogenic impacts on
- particulate organic carbon cycling in the North Sea (APOC)" (03F0874C). It is also supported by
- the Helmholtz research programme POF IV "The Changing Earth Sustaining our Future" on
- 343 "Topic 4: Coastal zones at a time of global change".

## 344 **Open Research**

- 345 Datasets for this research are available at zenodo https://doi.org/10.5281/zenodo.7560324. The
- source code of TOCMAIM with a coupling interface to 3-D physical-biogeochemical models are
- 347 stored in Mendeley Data with open access at http://dx.doi.org/10.17632/2vvny3xd85.2

# 348 References

- Amoroso, R. O., Pitcher, C. R., Rijnsdorp, A. D., McConnaughey, R. A., Parma, A. M., Suuronen,
- P., et al. (2018). Bottom trawl fishing footprints on the world's continental shelves.
- Proceedings of the National Academy of Sciences of the United States of America, 115(43),
- 352 E10275- E10282.
- Bergman, M.J.N., & Van Santbrink, J. W. (2000). Mortality in Megafaunal Benthic Populations
   Caused by Trawl Fisheries on the Dutch Continental Shelf in the North Sea in 1994. ICES. J.
   Mar. Sci. 57 (5), 1321–1331. doi: 10.1006/jmsc.2000.0917
- Bockelmann, F.D., Puls, W., Kleeberg, U., Müller, D., & Emeis, K.C. (2018). Mapping mud content and median grain-size of North Sea sediments – A geostatistical approach. Marine Geology
- 358 **397, 60–71.** https://doi.org/10.1016/j.margeo.2017.11.003.
- Clark, R. A., & Frid, C. L. J. (2001). Long-term changes in the North Sea ecosystem.
- 360 Environmental Reviews. 9(3): 131-187. https://doi.org/10.1139/a01-005
- Daewel, U., & Schrum, C. (2013). Simulating long term dynamics of the coupled North Sea and
   Baltic Sea ecosystem with ECOSMO II: Model description and validation. Journal of Marine
   Systems, 119–120, 30–49. https://doi.org/10.1016/j.jmarsys.2013.03.008
- 364 De Borger, E., Tiano, J., Braeckman, U., Rijnsdorp, A. D., & Soetaert, K. (2021). Impact of bottom 365 trawling on sediment biogeochemistry: A modelling approach. Biogeosciences, 18, 2539–
- 366 **2557**. <u>https://doi.org/10.5194/bg-2020-328</u>
- de Haas, H., Boer, W., & van Weering, T. C. E. (1997). Recent sedimentation and organic carbon
   burial in a shelf sea: the North Sea, Mar. Geol., 144, 131–146,
- 369 https://doi.org/10.1016/S0025-3227(97)00082-0
- de Wilde, P. A. W. J., & Kok, E. M. B. A. (1986). Biomass and activity of benthic fauna on the
  fladen ground (northern North Sea). Netherlands Journal of Sea Research, 20 (2-3), 313-323.
  https://doi.org/10.1016/0077-7579(86)90053-0
- 373 Diesing, M., Thorsnes, T., & Bjarnadóttir, L. R. (2021). Organic carbon densities and
- 374 accumulation rates in surface sediments of the North Sea and Skagerrak, Biogeosciences,
- 375 18, 2139–2160, https://doi.org/10.5194/bg-18-2139-2021

- 376 Dounas, C., Davies, I., Triantafyllou, G., Koulouri, P., Petihakis, G., Arvanitidis, C., Sourlatzis, G.,
- 377 & Eleftheriou, A. (2007). Large-scale impacts of bottom trawling on shelf primary
- 378 productivity. Continental Shelf Research, 27(17), 2198–2210.
- 379 https://doi.org/10.1016/j.csr.2007.05.006

- Eigaard, O.R., Bastardie, F., Hintzen, N.T., Buhl-Mortensen, L., Buhl-Mortensen, P., Catarino, R.,
- et al. (2017). The footprint of bottom trawling in European waters: distribution, intensity,
- and seabed integrity. ICES J Mar Sci 74, 847–865. https://doi.org/10.1093/icesjms/fsw194.
  Engelhard, G. H. (2016). North Sea Historical Effort by Rectangle by Month: 1913 -1980. Cefas,
  - UK. V1. doi: https://doi.org/10.14466/CefasDataHub.24
- Epstein, G., Middelburg, J. J., Hawkins, J. P., Norris, C. R., & Roberts, C. M. (2022). The impact of
   mobile demersal fishing on carbon storage in seabed sediments. Global Change Biology, 28,
   2875–2894. https://doi.org/10.1111/gcb.16105
- Freitas, F. S., Pika, P. A., Kasten, S., Jørgensen, B. B., Rassmann, J., Rabouille, C., et al. (2021).
- 389 New insights into large-scale trends of apparent organic matter reactivity in marine
- sediments and patterns of benthic carbon transformation, Biogeosciences, 18, 4651–4679,
   https://doi.org/10.5194/bg-18-4651-2021
- Halpern, B. S., Walkbridge, S., Selkoe, K. A., Kappel, C. V., Micheli, F., D'Agrosa, C., et al. (2008).
  A global map of human impact on marine ecosystems. Science, 319, 948–953.
  https://doi.org/10.1126/science.1149345
- Hartnett, H., Keil, R., Hedges, J., & Devol, A. H. (1998). Influence of oxygen exposure time on
  organic carbon preservation in continental margin sediments. Nature 391, 572–575.
  https://doi.org/10.1038/35351
- Hiddink, J.G., Jennings, S., Sciberras, M., Szostek, C.L., Hughes, K.M., Ellis, N., et al. (2017).
- Global analysis of depletion and recovery of seabed biota after bottom trawling disturbance.
  Proc Natl Acad Sci U S A, 114(31):8301-8306. doi: 10.1073/pnas.1618858114.
- 401 ICES (2007). Structure and dynamics of the North Sea benthos. ICES Cooperative Research
   402 Reports (CRR). Report. https://doi.org/10.17895/ices.pub.5451
- 403 ICES (2019). OSPAR request 2018 for spatial data layers of fishing intensity/pressure. Data
   404 Outputs. <u>https://doi.org/10.17895/ices.data.4686</u>.
- ICES (2021). ICES 2021 Greater North Sea Ecoregion Fisheries Overview Data Output file. Data
   Outputs. <u>https://doi.org/10.17895/ices.data.9158</u>.
- Kaiser, M. J., Clarke, K. R., Hinz, H., Austen, M. C. V., Somerfield, P. J., & Karakassis, I. (2006).
  Global analysis of response and recovery of benthic biota to fishing. Marine Ecology
- 409 Progress Series, 311, 1–14. <u>https://doi.org/10.3354/meps311001</u>
- Keil, R. (2017). Anthropogenic forcing of carbonate and organic carbon preservation in marine
   sediments. Annual Review of Marine Science, 9(1), 151–172.
- 412 <u>https://doi.org/10.1146/annurev-marine-010816-060724</u>
- 413 Kroodsma, D.A., Mayorga, J., Hochberg, T., Miller, N.A., Boerder, K., Ferretti, F., et al. (2018).
- 414 Tracking the global footprint of fisheries. Science 359, 904–908.
- 415 https://doi.org/10.1126/science.aao5646.

| 416 | LaRowe, D. E., Arndt, S., Bradley, J. A., Estes, E. R., Hoarfrost, A., Lang, S. Q., et al. (2020). The   |
|-----|--|
| 417 | fate of organic carbon in marine sediments—New insights from recent data and analysis.                   |
| 418 | Earth-Science Reviews, 204, 103146. <u>https://doi.org/10.1016/j.earscirev.2020.103146</u>               |
| 419 | Martín, J., Puig, P., Palanques, A., Masqué, P., & García-Orellana, J. (2008). Effect of commercial      |
| 420 | trawling on the deep sedimentation in a Mediterranean submarine canyon. Marine Geology,                  |
| 421 | 252(3–4), 150–155. https://doi.org/10.1016/j.margeo.2008.03.012  |
| 422 | Middelburg, J. J. (2018). Reviews and syntheses: To the bottom of carbon processing at the               |
| 423 | seafloor. Biogeosciences, 15(2), 413–427. <u>https://doi.org/10.5194/bg-15-413-2018</u>                  |
| 424 | Oberle, F. K. J., Storlazzi, C. D., & Hanebuth, T.J.J. (2016). "What a drag: Quantifying the global      |
| 425 | impact of chronic bottom trawling on continental shelf sediment". Journal of Marine                      |
| 426 | Systems. 159: 109–119. Bibcode:2016JMS1591090. doi:10.1016/j.jmarsys.2015.12.007                         |
| 427 | O'Neill, F.G., Robertson, M., Summerbell, K., Breen, M., & Robinson, L.A. (2013). The                    |
| 428 | mobilisation of sediment and benthic infauna by scallop dredges. Marine Environmental                    |
| 429 | Research 90, 104–112. https://doi.org/10.1016/j.marenvres.2013.06.003.                                   |
| 430 | Paradis, S., Puig, P., Sanchez-Vidal, A., Masqué, P., Garcia-Orellana, J., Calafat, A., & Canals, M.     |
| 431 | (2018). Spatial distribution of sedimentation-rate increases in Blanes Canyon caused by                  |
| 432 | technification of bottom trawling fleet. Progress in Oceanography, 169, 241–252.                         |
| 433 | https://doi.org/10.1016/j.pocean.2018.07.001   |
| 434 | Paradis, S., Goñi, M., Masqué, P., Durán, R., Arjona-Camas, M., Palanques, A., & Puig, P. (2021).        |
| 435 | Persistence of biogeochemical alterations of deep-sea sediments by bottom trawling.                      |
| 436 | Geophysical Research Letters, 48(2), e2020GL091279.  |
| 437 | https://doi.org/10.1029/2020gl091279   |
| 438 | Pitcher, C. R., Hiddink, J. G., Jennings, S., Collie, J., Parma, A. M., Amoroso, R., et al. (2022).      |
| 439 | Trawl impacts on the relative status of biotic communities of seabed sedimentary habitats in             |
| 440 | 24 regions worldwide. Proceedings of the National Academy of Sciences, 119(2),                           |
| 441 | e2109449119. <u>https://doi.org/10.1073/pnas.2109449119</u>  |
| 442 | Rijnsdorp, A.D., Bolam, S.G., Garcia, C., Hiddink, J.G., Hintzen, N.T., van Denderen, P.D. & van         |
| 443 | Kooten, T. (2018). Estimating sensitivity of seabed habitats to disturbance by bottom                    |
| 444 | trawling based on the longevity of benthic fauna. Ecol Appl, 28: 1302-1312.                              |
| 445 | https://doi.org/10.1002/eap.1731   |
| 446 | Rosenberg, R., Hellman, B., & Lundberg, A. (1996). Benthic macrofaunal community structure in            |
| 447 | the Norwegian Trench, deep skaggerrak. Journal of Sea Research, 35 (1-3), 181-188.                       |
| 448 | https://doi.org/10.1016/S1385-1101(96)90745-5  |
| 449 | Sala, E., Mayorga, J., Bradley, D., Cabral, R. B., Atwood, T. B., Auber, A., et al. (2021). Protecting   |
| 450 | the global ocean for biodiversity, food and climate. Nature, 592, 397– 402.                              |
| 451 | <u>https://doi.org/10.1038/s41586-021-03371-z</u>  |
| 452 | Schrum, C. (1997). Thermohaline stratification and instabilities at tidal mixing fronts - Results of     |
| 453 | an eddy resolving model for the German Bight. Continental Shelf Research, 17, 689-716.                   |
| 454 | Sciberras, M., Hiddink, J. G., Jennings, S., Szostek, C. L., Hughes, K. M., Kneafsey, B., et al. (2018). |
| 455 | Response of benthic fauna to experimental bottom fishing: A global meta-analysis. Fish and               |
| 456 | Fisheries, 19(4), 698–715. https://doi.org/10.1111/faf.12283   |
|     |  |

Ståhl, H., Tengberg, A., Brunnegård, J., Bjørnbom, E., Forbes, T. L., Josefson, A. B., et al. (2004).
 Factors influencing organic carbon recycling and burial in Skagerrak sediments. *Journal of Marine Research*, 62, 867-907.

Thomas, H., Bozec, Y., de Baar, H. J. W., Elkalay, K., Frankig-noulle, M., Schiettecatte, L.-S., et al.
(2005). The carbon budget of the North Sea. Biogeosciences, 2(1), 87–96.

462 https://doi.org/10.5194/bg-2-87-2005

Thurstan, R., Brockington, S. & Roberts, C. (2010). The effects of 118 years of industrial fishing
on UK bottom trawl fisheries. Nat Commun 1, 15. https://doi.org/10.1038/ncomms1013

Tiano, J. C., Depestele, J., Van Hoey, G., Fernandes, J., van Rijswijk, P., & Soetaert, K. (2022).

- 466 Trawling effects on biogeochemical processes are mediated by fauna in high-energy
   467 biogenic-reef-inhabited coastal sediments, Biogeosciences, 19, 2583–2598,
   468 https://doi.org/10.5194/bg-19-2583-2022
- Tillin, H. M., Hiddink, J. G., Jennings, S., & Kaiser, M. J. (2006). Chronic bottom trawling alters
   the functional composition of benthic invertebrate communities on a sea-basin scale.

471 Marine Ecology Progress Series, 318, 31–45. https://doi.org/10.3354/meps318031

- Zhang, W., & Wirtz, K. (2017). Mutual dependence between sedimentary organic carbon and
  infaunal macrobenthos resolved by mechanistic modeling. Journal of Geophysical Research:
  Biogeosciences, 122, 2509–2526. https://doi.org/10.1002/2017JG003909.
- Zhang, W., Wirtz, K., Daewel, U., Wrede, A., Kröncke, I., Kuhn, G., et al. (2019). The budget of
  macrobenthic reworked organic carbon A modelling case study of the North Sea. Journal
- of Geophysical Research: Biogeosciences, 124(6), 1446–1471.
- 478 https://doi.org/10.1029/2019JG005109
- Zhang, W., Neumann, A., Daewel, U., Wirtz, K., van Beusekom, J.E.E., Eisele, A., et al. (2021).

480 Quantifying importance of macrobenthos for benthic-pelagic coupling in a temperate

481 coastal shelf sea. Journal of Geophysical Research: Oceans, 126, e2020JC016995,

482 doi:10.1029/2020JC016995

- Zonneveld, K. A. F., Versteegh, G. J. M., Kasten, S., Eglinton, T. I., Emeis, K.-C., Huguet, C., et al.
- 484 (2010). Selective preservation of organic matter in marine environments; processes and
- impact on the sedimentary record, Biogeosciences, 7, 483–511, https://doi.org/10.5194/bg7-483-2010

# 487 References from the Supporting Information

Eigaard, O.R., Bastardie, F., Breen, M., Dinesen, G.E., Hintzen, N.T., Laffargue, P., et al. (2016).

Estimating seabed pressure from demersal trawls, seines, and dredges based on gear design and dimensions. ICES J. Mar. Sci. 73, i27-i43. https://doi.org/10.1093/icesjms/fsv099.

- 491 Hedges, J.I., Baldock, J.A., Gélinas, Y., Lee, C., Peterson, M.L., Wakeham, S.G. (2002). The
- 492 biochemical and elemental compositions of marine plankton: a NMR perspective. Mar Chem493 78:47–63.

- 494 Lorenzen C. J. (1967), Determination of chlorophyll and pheo-pigments: spectrophotometric
- 495 equations. Limnology and Oceanography 12, issue 2, pages 343-346, doi:
- 496 **10.4319/lo.1967.12.2.0343**
- 497 O'Neill, F. G. & Summerbell, K. (2011). The mobilisation of sediment by demersal otter trawls.
- 498 Mar. Pollut. Bull., 62, pp. 1088-1097, 10.1016/j.marpolbul.2011.01.038